Author's response to reviews

Title: Transfusion Indication Threshold Reduction (TITRe2) randomised controlled trial in cardiac surgery: statistical analysis plan

Authors:

Katie Pike (katie.pike@bristol.ac.uk)
Rachel L Nash (rachel.nash@bristol.ac.uk)
Gavin J Murphy (gjm19@leicester.ac.uk)
Barnaby C Reeves (barney.reeves@bristol.ac.uk)
Chris A Rogers (chris.rogers@bristol.ac.uk)

Version: 3
Date: 8 December 2014

Author's response to reviews: see over
08 December 2014

Professors D Altman, C Furberg and J Grimshaw
Editors in chief of Trials

Dear Editors,

Thank you for your further comments regarding our manuscript MS 8771326411211248, entitled “Transfusion Indication Threshold Reduction (TITRe2) randomised controlled trial in cardiac surgery: statistical analysis plan”.

As requested, we have revised the manuscript to add the name of the NHS Research Ethics Committee, and to ensure that our manuscript conforms to the journal style.

We hope that the revised manuscript will meet with the Editors’ approval.

Yours sincerely,

Dr Chris Rogers